Author’s response to reviews

Title: Maternal care practices among the ultra poor households in rural Bangladesh: a qualitative exploratory study

Authors:

Nuzhat Choudhury (nuzhat.choudhury@yahoo.com)
Syed Masud Ahmed (ahmed.sm@brac.net)

Version: 5 Date: 27 February 2011

Author’s response to reviews:

Dear Editor,

Thank you very much to accept the manuscript for publication in BMC Pregnancy and Childbirth.

We have made corrections as suggested by Editorial Production Team.

Thank you again

Sincerely,

Nuzhat Choudhury